Author's response to reviews

Title: Ochronosis as an unusual cause of a valvular defect: a case report

Authors:

Andreas Wilke (dr.andreas.wilke@googlemail.com)
Dietmar Steverding (dsteverding@hotmail.com)

Version: 3 Date: 27 May 2008

Author's response to reviews:

Dear Sir or Madam,

Please find enclosed a revised version of our manuscript entitled "Ochronosis as an unusual cause of a valvular defect: a case report". As advised by the editorial team, we have rephrase "figure legend" as "figure legends" and have removed the marginal numbers from the figures.

I look forward to hearing from you.

Yours sincerely,
Dietmar Steverding

Priv.-Doz. Dr. Dietmar Steverding
BioMedical Research Centre
School of Medicine, Health Policy and Practice
University of East Anglia
Norwich NR4 7TJ
Tel.: 0044-1603-591291
E-mail: d.steverding@uea.ac.uk